Author’s response to reviews

Title: Salvage living donor liver transplantation after percutaneous transluminal angioplasty for recurrent Budd-Chiari syndrome: a case report

Authors:

Yusaku Shirai (shirai@narmed-u.ac.jp)
Hitoshi Yoshiji (yoshijih@narmed-u.ac.jp)
Saiho Ko (ko@narmed-u.ac.jp)
Masaahiru Yamazaki (yamazaki@narmed-u.ac.jp)
Yasuhide Ikenaka (ikenaka@narmed-u.ac.jp)
Ryuichi Noguchi (rnoguchi@narmed-u.ac.jp)
Chie Morioka (morioka@narmed-u.ac.jp)
Kosuke Kaji (kaji@narmed-u.ac.jp)
Yosuke Aihara (aihara@narmed-u.ac.jp)
Keisuke Nakanishi (nakanishi@narmed-u.ac.jp)
Junichi Yamao (juny@narmed-u.ac.jp)
Masahisa Toyohara (msat@narmed-u.ac.jp)
Akira Mitoro (akira@narmed-u.ac.jp)
Masayoshi Sawai (masay@narmed-u.ac.jp)
Motoyuki Yoshida (motoyu@narmed-u.ac.jp)
Masao Fujimoto (masaf@narmed-u.ac.jp)
Masahito Uemura (masahi@narmed-u.ac.jp)
Yoshiyuki Nakajima (yoshiyn@narmed-u.ac.jp)
Hiroshi Fukui (hfuki@narmed-u.ac.jp)

Version: 4 Date: 27 January 2011

Author’s response to reviews: see over
Dear Editor:

Re: MS1639632092424967

We appreciate the reviewer’s comments concerning our manuscript entitled “Salvage living donor liver transplantation after percutaneous transluminal angioplasty for recurrent Budd-Chiari syndrome: a case report”.

According to editor’s comment, we add the Consent section and include the ethnicity of the patient (Japanese) in the re-revised manuscript. Also, attached the file of written informed consent form.

We hope that our re-revised manuscript will be now acceptable for publication in the Journal of Medical Case Reports.

Thank you very much in advance for your time and effort.

Sincerely yours,

Hitoshi Yoshiji, M.D., Ph.D.
The Third Department of Internal Medicine,
Nara Medical University,
840 Shijo-cho, Kashihara, Nara 634-8522,
Japan.
E-mail: yoshijih@naramed-u.ac.jp
TEL: 81-744-22-3051 (ext. 2314)
FAX: 81-744-24-7122

Enclosure(s)